Author’s response to reviews

Title: Depression among HIV positive pregnant women in Zimbabwe: a primary health care based cross-sectional study

Authors:

Eugenia Nyamukoho (eugenianyamukoho@yahoo.com)
Walter Mangezi (wmangezi@yahoo.co.uk)
Bazondlile Dube (bazoedube@gmail.com)
Ruth Verhey (ruth.verhey@zol.co.zw)
Dixon Chibanda (dichi@zol.co.zw)

Version: 2 Date: 27 Dec 2018

Author’s response to reviews:

The Editor

RE: Depression among HIV positive pregnant women in Zimbabwe: a primary health care based cross-sectional study

Thank you for giving us this opportunity to respond to the minor revisions suggested below. We have addressed all the suggestions as follows:

1. Please include a title page to your manuscript that is properly formatted according to our submission guidelines (https://bmcpregnancychildbirth.biomedcentral.com/submission-guidelines/preparing-your-manuscript). The title page should: A title page has been included.

   - present the title that includes, if appropriate, the study design

   - list the full names, institutional addresses and email addresses for ALL authors

   - indicate the corresponding author: This has been carried out as requested

2. Please format your abstract and the main manuscript to include the following subheadings: Background, Methods, Results, Conclusions. The suggested subheadings have been included
3. In the Funding section, please also state the role of the funding body in the design of the study, collection, analysis, interpretation of data, and in writing the manuscript. The role of the funding has been included which was to support the last author otherwise the funder was not involved in the actual study.

4. Please add a section "Consent for publication" in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript. This has been included in the section on declaration.

5. Please upload English language versions of the socio-demographic questionnaire developed specifically for use in this study as a supplementary file. If the questionnaire is not yours or is already published by you, please supply reference and/or link to it. This has been done.

6. Please proofread your manuscript, ensure that the paragraphs are correctly formatted, remove any unnecessary spacing, and ensure that your revised submission is in the final form for publication. Please also remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only. We have read the manuscript through again and removed all track changes.

We hope these responses will meet your satisfaction.

Thank you

Dr. D Chibanda